Author's response to reviews

Title: Association between smoking and chronic kidney disease: a case control study

Authors:

Rabi Yacoub (Rabiyaco@buffalo.edu)
Habib Habib (Hhabib@smmcnj.org)
Ayham Lahdo (Ayham_lahdo@hotmail.com)
Radwan Al Ali (Radgali@gmail.com)
Leon Varjabedian (leonvarj@buffalo.edu)
George Atalla (Gaa82@hotmail.com)
Nader Kassis Akl (Naderkassis@yahoo.com)
Saleem Aldakheel (Salimdakhil@yahoo.com)
Saeed Alahdab (Saeedalahdab@yahoo.com)
Sami Albitar (Samialbitar@hotmail.com)

Version: 6 Date: 25 November 2010

Author's response to reviews:

Q: References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].
A: Corrected, EndNote numbered formate used.